Author's response to reviews

Title: Invasive fungal sinusitis in patients with hematological malignancy: 15 years experience in a single university hospital in Taiwan

Authors:

Chien-Yuan Chen (chienyuan@ntu.edu.tw)
Wang-Huei Sheng (whsheng@ntu.edu.tw)
Aristine A Cheng (arischeng@ntuh.gov.tw)
Yee-Chun Chen (yeechunchen@ntuh.gov.tw)
Woei Tsay (woei@ntuh.gov.tw)
Jih-Luh Tang (tangjl@ntuh.gov.tw)
Shang-Yi Huang (syhuang55@ntuh.gov.tw)
Shan-Chwen Chang (scchang@ntuh.gov.tw)
Hwei-Fang Tien (hftien@ntuh.gov.tw)

Version: 4 Date: 12 September 2011

Author's response to reviews:

Roxane Rajabi, MD, PhD
Editor-in-Chief
BMC Infectious Diseases
RE: MS 8804397255239615
Sep 12, 2011
Dear Dr. Rajabi:
Thank you very much for your kind attention to our paper entitled “Invasive fungal sinusitis in patients with hematological malignancy: 15 years experience in a single university hospital in Taiwan.” by CY Chen et al. We also sincerely appreciate your acceptance for our manuscript. We have revised our manuscript according to the comments and suggestions of the editor and the reviewers. What we have done are listed and explained in the following:

Major revisions (we require the author to make these changes)

Affiliations: Each affiliation should be written in full, in the following format: Department, Institute, (Street), City, (Postal code), Country. This may mean repeating institutional addresses if the authors are from different departments within an institution
Response: Thank you very much, we have revised this information.

Tables: Table 4 has not been cited in the text please either cite the table or remove it from the manuscript.
Response: Thank you! We cited Table 4 in Page 14, line 8.

Minor revisions (we can make these changes for you, although it will speed up
publication of your manuscript if you do them while making the major changes above)
Underlining: Please remove underlining
Response: Thank you! We removed the underlining.
Background: Please change the title "Introduction" to "Background"
Response: Thank you! We changed Introduction to Background

We appreciate those comments and suggestions of the editor and reviewer to improve the quality of our manuscripts. Thank you very much for your kind attention to our paper.

Sincerely yours,

Wang-Huei Sheng, M.D., Ph.D..
Division of Infectious Diseases,
Department of Internal Medicine,
National Taiwan University Hospital,
7 Chung-Shan South Road, Taipei 100, Taiwan
Email: whsheng@ntu.edu.tw
Tel: 886-2-23123456-66842
Fax: 886-2-23971412